Author's response to reviews

Title: Retention of foreign body in the gastrointestinal tract: a sign of congenital obstructive anomaly

Authors:

Pravas C Subudhi (beam.diagnostics@gmail.com)
Shivaram P Singh (sudhasingh@sify.com)
Chudamani Meher (beam.diagnostics@gmail.com)
Omprakash Agrawal (beam.diagnostics@gmail.com)

Version: 3 Date: 29 February 2008

Author's response to reviews: see over
The Editor,
The Journal of Medical Case Reports

Dear Editor,

We submit this revised manuscript entitled “Retention of foreign body in the gut can be a sign of congenital obstructive anomaly: a case report” for publication in your journal. The revision has been made only with regard to formatting of article as per your editorial instructions as prescribed the journal. No changes have been made in the main text of the article. We believe the paper may be of particular interest to the readers of your journal as it would remind them of the need to keep in mind “congenital obstructive lesions” in cases of foreign body retention in the gut. This case report has not been submitted elsewhere for publication in whole or in part, and all the authors listed have approved the manuscript. None of the authors has any potential financial conflict of interest related to this manuscript. We have read and have completely abided by the statement of ethical standards and the policy for manuscripts submitted to JMCR.

Thank you very much for your attention and consideration. Looking forward to your decision with kind personal regards.

Sincerely,

Shivaram Prasad Singh, MBBS, MD, DM [Vellore]
Associate Professor & Head,
Department of Gastroenterology
S.C.B. Medical College,
Cuttack, India
Tel: +91-671-2605466
Fax: +91-671-2605466
Email: sudhasingh@sify.com